Review

Clinical Benefits and Risks of Antiamyloid Antibodies in Sporadic Alzheimer Disease: Systematic Review and Network Meta-Analysis With a Web Application

Danko Jeremic, PharmD, PhD; Juan D Navarro-Lopez*, PharmD, PhD; Lydia Jimenez-Diaz*, PharmD, PhD

Neurophysiology & Behavior Lab, Institute of Biomedicine (IB-UCLM) and Instituto de Investigación Sanitaria de Castilla-La Mancha (IDISCAM), Faculty of Medicine of Ciudad Real, University of Castilla-La Mancha, Ciudad Real, Spain *these authors contributed equally

Corresponding Author:

Lydia Jimenez-Diaz, PharmD, PhD Neurophysiology & Behavior Lab Institute of Biomedicine (IB-UCLM) and Instituto de Investigación Sanitaria de Castilla-La Mancha (IDISCAM), Faculty of Medicine of Ciudad Real University of Castilla-La Mancha Camino Moledores s/n Ciudad Real, 13071 Spain Phone: 34 926295300 Email: lydia.jimenez@uclm.es

Abstract

Background: Despite the increasing approval of antiamyloid antibodies for Alzheimer disease (AD), their clinical relevance and risk-benefit profile remain uncertain. The heterogeneity of AD and the limited availability of long-term clinical data make it difficult to establish a clear rationale for selecting one treatment over another.

Objective: The aim of this work was to assess and compare the efficacy and safety of antiamyloid antibodies through an interactive online meta-analytic approach by performing conventional pair-wise meta-analyses and frequentist and Bayesian network meta-analyses of phase II and III clinical trial results. To achieve this, we developed AlzMeta.app 2.0, a freely accessible web application that enables researchers and clinicians to evaluate the relative and absolute risks and benefits of these therapies in real time, incorporating different prior choices and assumptions of baseline risks of disease progression and adverse events.

Methods: We adhered to PRISMA-NMA (Preferred Reporting Items for Systematic Reviews and Meta-Analyses extension for reporting of systematic reviews with network meta-analysis) and GRADE (Grading of Recommendations, Assessment, Development, and Evaluation) guidelines for reporting and rating the certainty of evidence. Clinical trial reports (until September 30, 2024) were retrieved from PubMed, Google Scholar, and clinical trial databases (including ClinicalTrials.gov). Studies with <20 sporadic AD patients and a modified Jadad score <3 were excluded. Risk of bias was assessed with the RoB-2 tool. Relative risks and benefits have been expressed as risk ratios and standardized mean differences, with confidence, credible, and prediction intervals calculated for all outcomes. For significant results, the intervention effects were ranked in frequentist and Bayesian frameworks, and their clinical relevance was determined by the absolute risk per 1000 people and number needed to treat (NNT) for a wide range of control responses.

Results: Among 7 treatments tested in 21,236 patients (26 studies with low risk of bias or with some concerns), donanemab was the best-ranked treatment on cognitive and functional measures, and it was almost 2 times more effective than aducanumab and lecanemab and significantly more beneficial than other treatments on the global (cognitive and functional) Clinical Dementia Rating Scale-Sum of Boxes (NNT=10, 95% CI 8-16). Special caution is required regarding cerebral edema and microbleeding due to the clinically relevant risks of edema for donanemab (NNT=8, 95% CI 5-16), aducanumab (NNT=10, 95% CI 6-17), and lecanemab (NNT=14, 95% CI 7-31), which may outweigh the benefits.

Conclusions: Our results showed that donanemab is more effective and has a safety profile similar to aducanumab and lecanemab, highlighting the need for treatment options with improved safety. Potential bias may have been introduced in the included trials due to unblinding caused by frequent cerebral edema and microbleeds, as well as the impact of the COVID-19 pandemic.

(J Med Internet Res 2025;27:e68454) doi: 10.2196/68454

KEYWORDS

Alzheimer disease; antibodies; donanemab; aducanumab; lecanemab

Introduction

Until lately, the only available therapies for Alzheimer disease (AD) were related to symptomatic presentation, and this situation remains in most countries. Regulatory approvals of aducanumab (2021), lecanemab (2023), and donanemab (2024) in the United States have marked a new era in AD treatment, as these antibodies demonstrated the potential to modify disease progression by targeting amyloid β (A β), which involves toxic peptides deemed crucial in the AD pathophysiology [1]. While the development of aducanumab was discontinued earlier in 2024 (January 2024) [2], lecanemab and donanemab are currently under review for approval in Europe and worldwide [3], and lecanemab is already available for patients outside of the United States, including those in the United Kingdom, Japan, China, South Korea, Israel, and the United Arab Emirates [4-6].

Despite the clear advantages of anti-A β antibodies, substantial doubts remain about their risk-to-benefit profile and clinical relevance, and they still have to demonstrate clinically meaningful effects in real life [7-10]. There is limited understanding of their long-term effects, and the safety profile has yet to be fully explored in broader populations. Although reductions in amyloid load are well-documented and substantial, it remains uncertain whether these changes lead to significant improvements in cognition, disease progression, and daily functioning. The complexity of AD and its varying pathophysiological characteristics make it difficult to predict how these therapies will impact different patient subgroups, including those with different levels of tau pathology, disease stages, or varying risks of cognitive/functional decline and adverse events (AEs) [7,11-13].

We previously performed a web application meta-analysis of these treatments in phase III randomized placebo-controlled clinical trials (RCTs) involving sporadic AD [14]. Our analysis included aducanumab [15], lecanemab [16], bapineuzumab [17,18], and solanezumab [19,20], showing that aducanumab and lecanemab produced the most promising cognitive, functional, and biomarker results. However, these effects were achieved at the great expense of increased AEs, primarily amyloid-related imaging abnormalities (ARIAs) in the form of vasogenic cerebral edema and sulcal effusion (ARIA-E) and microhemorrhage and superficial siderosis (ARIA-H) [14]. Our study had some limitations as we analyzed only large phase III RCTs (>200 patients) with the conventional (pair-wise) meta-analysis method. This did not allow us to compare the effects of the treatments and rank their performance, and we failed to include smaller studies [21,22] and novel antibodies, including donanemab [23,24], gantenerumab [25,26], and crenezumab [27].

Unlike conventional pair-wise meta-analysis, which only compares treatments that have been directly tested against each other in clinical trials, a network meta-analysis (NMA) allows

```
https://www.jmir.org/2025/1/e68454
```

for indirect treatment comparisons. This means that even if 2 interventions have not been directly compared in a trial, a NMA can estimate their relative effectiveness based on their shared comparability to a common reference intervention. Therefore, the aim of this study was to update and expand our previous application [14] in 3 steps. First, we performed a conventional (frequentist) random effects meta-analysis with the DerSimonian and Laird method [28,29] on an extended dataset and additional outcomes in sporadic AD. Then, we conducted a frequentist [30] and Bayesian NMA [31] of phase II and III RCTs in order to compare the effectiveness and safety of anti-A β antibodies. Finally, we evaluated the clinical relevance of these interventions by estimating the absolute benefits and risks over an entire range of placebo responses and different prior assumptions of heterogeneity.

Methods

Search Strategy, Quality Assessments, and Inclusion Criteria

We followed the PRISMA-NMA (Preferred Reporting Items for Systematic Reviews and Meta-Analyses extension for reporting of systematic reviews with network meta-analysis) guidelines (Multimedia Appendix 1) [32] and GRADE (Grading of Recommendations, Assessment, Development, and Evaluation) guidelines [33] for reporting and rating the certainty (quality) of underlying evidence. Primary study sources were PubMed, Google Scholar, and multiple clinical trial databases, including the US Clinical Trials Registry (ClinicalTrials.gov), the EU Clinical Trials Registry, the Australian New Zealand Clinical Trials Registry (ANZCTR), and the International Clinical Trials Registry Platform (ICTRP) of the World Health Organization. The search terms were as follows: "Alzheimer's," "sporadic," "mild cognitive impairment," "phase 3," "phase 2," "monoclonal antibody," "passive immunotherapy," "Aducanumab," "BIIB037," "Gantenerumab," "Lecanemab," "BAN-2401," "Solanezumab," "LY2062430," "Crenezumab," "AAB-001," "Donanemab," "Bapineuzumab," and "LY3002813." No age or language restrictions were applied. We excluded studies that (1) tested <20 sporadic AD patients and (2) were not phase II/III RCTs with a report quality of ≥ 3 on the modified Jadad scale. The details of the electronic search strategy and modified Jadad scale assessments can be found in Multimedia Appendix 2. In brief, apart from the regular examination of study randomization, blinding, dropouts, and withdrawals [34], we evaluated whether there were clear descriptions of the inclusion and exclusion criteria, statistical analysis, and methods used to assess AEs such as ARIA. The search for eligible studies was performed by each author independently (until September 30, 2024), and any disagreements were resolved through consensus. Excel was then used to automatically eliminate duplicate studies. The study did not include a registered review protocol, and it mainly expanded our previously published protocol [14] that was initially limited

in scope, focusing only on larger phase III studies with no less than 200 patients in each arm.

Primary (Cognitive and Functional) Outcomes

The primary outcomes included cognitive and functional measures from all or most studies, and they involved mean changes from baseline in the (1) AD Assessment Scale-Cognitive Subscale (ADAS-Cog), (2) Mini-Mental State Examination (MMSE), and (3) Clinical Dementia Rating Scale-Sum of Boxes (CDR-SB). The results from the primary outcomes were evaluated to assess whether they achieved the minimal clinically important differences (MCIDs) obtained in previous studies [13,14,35].

Secondary (Biomarker and Safety) Outcomes

The secondary outcomes were biomarker and safety measures reported by all or most trials. The biomarkers included mean changes from baseline in the amyloid burden on positron emission tomography (PET; centiloids) and the cerebrospinal fluid (CSF) biomarkers of A $\beta_{1.42}$ and p-tau-(Thr₁₈₁). Safety outcomes were serious AEs, tolerability (treatment discontinuation due to AEs), and total events of ARIA-E (cerebral edema and sulcal effusion), ARIA-H (cerebral microhemorrhage and superficial siderosis), headaches, dizziness, falls, arthralgia, diarrhea, urinary infection, and nasopharyngitis.

Tertiary Outcomes and Additional Measures

Tertiary outcomes included AEs reported by few studies only: total events of ARIA-E in *APOE*- ε 4 carriers and noncarriers, fatigue, nausea, back pain, and upper respiratory infections.

Apart from the outcome measures, we extracted the inclusion criteria and baseline/participant characteristics of the primary studies, including mean age (years, SD), sex/ethnicity/race, *APOE* status, baseline ADAS-Cog, MMSE, and CDR-SB scores, dosage, and administration routes.

Data Analysis, Reporting, and Web Application Development

As in our previous study [14], in order to properly manage the differences in study-specific and patient-specific characteristics within the subgroups of the included studies, we followed the recommendation to treat each subgroup as a separate study whenever possible [36]. If it was not feasible, the subgroups from the same study were combined (pooled) [37] in order to avoid correlated effect sizes and double-counting [38-40].

Since some interventions were tested in different dose regimens, the main pair-wise meta-analysis (according to our previous methodology [14]), and frequentist and Bayesian NMAs were performed by pooling data from high-dose and low-dose regimens. Simultaneously, 2 sensitivity analyses were performed: one including only the high-dose regimen and the other including the low-dose regimen, while keeping other drugs in a single-dose regimen. If multiple low-dose regimens were available, the clinically most relevant one (with a higher dose) was selected.

The NMA results have been reported as frequentist and Bayesian estimates of the relative and absolute benefits and risks of

```
https://www.jmir.org/2025/1/e68454
```

interventions. Any discrepancy in the findings between the 2 NMA methods has been highlighted and explained in a GRADE summary of findings. Relative measures included standardized mean differences (SMDs) for benefits and risk ratios (RRs, lnRRs) for AEs. For statistically significant results with no excessive heterogeneity, relative measures have been converted to absolute risks (ARs) and benefits, expressed as AR per 1000 people [39] and numbers needed to treat (NNT) for additional harmful (NNTH, higher is better) [12,41-43] and beneficial outcomes (NNTB, lower is better) [37,41,42]. In AlzMeta.app 2.0, these calculations are allowed for a wide range of assumptions concerning the risks of decline and AEs in the control group (with no treatment). This has been implemented as it can be important for calculating the benefits and risks for patient populations with different expectations of baseline risks, based on prior history and clinical expertise in each particular case.

The web application AlzMeta.app 2.0 [44] has been mainly built in RStudio 2024.09 (R 4.4.1) by modifying and expanding the previous instance of the application, relying on key packages such as *shiny*, *tidyverse*, *dmetar*, *metafor*, *ggplot2*, *netmeta*, *rjags*, and *gemtc*. Multimedia Appendix 2 provides a detailed overview of R-based statistical packages used for data analysis (explained further) as well as quality assessments (risk of bias and modified Jadad scale). CSS and HTML have been used for further styling of the graphical interface.

Frequentist NMA

In the frequentist NMA, SMDs and RRs have been calculated together with their 95% CIs [30] and 95% prediction intervals (PIs) [45], and the results have been presented via forest plots and tables that separate direct and indirect estimates. The interventions have been ranked by P-score [46], representing the mean extent of certainty that a treatment is better than the competing treatments. P-score can have a value between 0 and 1 and is equivalent to the surface under the cumulative ranking curve (SUCRA) score in the Bayesian framework. A higher P-score indicates more effective or safer treatment when compared to all other treatments in the network, averaged over all competing treatments.

The values of I^2 statistics (95% CI) and their statistical significance were evaluated for overall heterogeneity or inconsistency in the network, which was then split into within-design (heterogeneity) and between-design (inconsistency), and the *P*-values of within-design Q-statistics have been reported. We followed the recommendations [36,47-50] to not include surrogate (biomarker) outcomes and findings with significant and excessive heterogeneity into the summary of findings, intervention rankings, and estimations of absolute effects.

Bayesian NMA

The Bayesian random effects NMA was performed in the Bayesian hierarchical model with Markov Chain Monte Carlo (MCMC) estimation [51]. Model parameters included 10 thinning iterations and 100,000 simulation iterations. The convergence of the MCMC model was assessed through Gelman-Rubin plots and the potential scale reduction factor.

Trace and density plots were used to decide the optimal parameters of the number of burn-in iterations, actual simulation iterations, and thinning parameters.

The relative effects (SMDs and RRs) have been presented in a forest plot for each outcome. For clinically relevant results achieved by multiple interventions, ranking was performed by SUCRA scores based on posterior probabilities in the Bayesian framework [52]. SUCRA scores range from 0% to 100% (inclusive) and can be interpreted as the estimated proportion of treatments worse than a given treatment. Values close to 100% suggest greater probability that a given treatment is the best. More information can be found in the documentation section in AlzMeta.app 2.0 [44].

The Bayesian NMA was performed by 3 different choices of heterogeneity priors in both the main and sensitivity analyses. First, we used uninformative priors (with large variance) and half-normal priors for SD, with scale factors based on overall heterogeneity measured in the frequentist NMA. Then, the estimates were obtained by half-normal priors assuming 3 times greater heterogeneity than measured. All these calculations are available for both the relative (SMDs and RRs) and absolute estimates (NNTBs, NNTHs, and ARs) in AlzMeta.app 2.0, and the corresponding code can be found in Multimedia Appendix 2.

The Bayesian network meta-regression was performed to check whether specific study characteristics influenced the effect size estimates for the primary outcomes, including the impact of the number of study sites, study duration (weeks), mean age of participants, and baseline ADAS-Cog score.

Rank Robustness

The robustness of SUCRA ranks was determined by excluding one study at a time in the leave-one-out session and calculating (1) the level of agreement beyond chance (interrater reliability) via quadratic weighted Cohen kappa [53], (2) how many (%) studies did not change any treatment rank, and (3) how many (%) studies displaced treatment ranks. The Cohen kappa values are often classified into levels of agreement for interpretation, for example, poor (<0%), slight (0%-20%), fair (21%-40%), moderate (41%-60%), substantial (61%-80%), and almost perfect (81%-100%) agreement. However, it should be mentioned that this is an ad hoc procedure and different interpretations are available [54], as it depends on the context and specific requirements for reliability. Even values above 90% do not necessarily indicate almost perfect agreement [53,54].

Rank Uncertainty

The uncertainty of SUCRA ranks has been visualized by rankograms [55], which show the distribution of ranking probabilities for each treatment to be the *n*th best option. Furthermore, we calculated Shannon normalized (information) entropy associated with SUCRA ranking probabilities, as proposed by Wu et al [56]. The authors proposed a very intuitive way to estimate uncertainty associated with ranking probabilities in NMA by applying the Shannon information entropy formula to obtain a normalized entropy score. They showed that the normalized entropy score provides a more accurate assessment

```
https://www.jmir.org/2025/1/e68454
```

of ranking uncertainty and does not depend on the number of analyzed treatments. This approach can be used to compare the uncertainty of treatment rankings within a NMA and also between different NMAs, which is crucial for the interpretation of results.

For a NMA, the most precise scenario would be absolute certainty in the ranking of treatments in our network. Therefore, each treatment would have 100% probability of being in one ranking position and 0% probability for the other positions (peaked distribution). Under this scenario, the entropy is zero bit, and normalized entropy equals zero. On the other hand, the normalized entropy reaches 1 in the least precise scenario when the ranking probabilities are the same for each rank (flat distribution).

Correlations

The correlation calculated among the cognitive, functional, and biomarker outcomes was the Pearson product moment correlation coefficient (r). For safety measures, Pearson r was computed only for data not significantly different from normal distribution, as determined by the Shapiro-Wilk test. The Spearman rank correlation coefficient was used for data not following normal distribution.

Influence and Sensitivity Analyses

Influential studies were identified initially by standard meta-analytic approaches: heterogeneity assessments, Baujat plots, influence diagnostics, and leave-one-out approach. Then, robustness was further evaluated by machine learning methods within graphic display of study heterogeneity (GOSH) plot analysis, in which the meta-analysis model was fit to all possible subsets of included studies [57,58]. Sensitivity analyses were then performed upon excluding detected influential observations. Furthermore, in frequentist NMA, the influence of each study on the treatment estimates was estimated within the main and sensitivity analyses by assessing the proportional contributions of direct comparisons [59] and the statistical importance of each study measured by the reduction of precision if removed from the analysis [60].

Risk of Bias and Publication Bias

The risk of bias at the study, outcome, and comparison levels was classified as "low," "raising some concerns," or "high," by using the revised Cochrane risk of bias (RoB-2) approach. Summary and study-level risk of bias was visualized via the RoB-2 web application [61,62], and the risk of bias at the outcome and comparison levels has been presented within the GRADE summary of findings tables. Publication bias was statistically evaluated for continuous outcomes with >10 studies [63] using the Egger regression test [37,64,65], and funnel plots were created using the Duval and Tweedie trim-and-fill procedure (Multimedia Appendix 2).

Certainty of Evidence and Impact of COVID-19

The certainty of underlying evidence at the comparison level within NMAs was assessed with rigorous adherence to the GRADE approach by focusing on within-study bias, reporting bias, indirectness, imprecision, heterogeneity, and incoherence [33]. The impact of the COVID-19 pandemic was assessed by

evaluating (1) the number (%) of affected patients per arm, (2) whether the impact of COVID-19 was analyzed in the primary study, and (3) how the pandemic affected each trial stage, AEs, unblinding, dropouts, delays, and missing doses.

Role of Funding Source

The funder of the study had no role in study design, data collection, analysis, interpretation, or writing of the report.

Ethical Considerations

This study did not require ethics board review as it did not involve human participants, identifiable patient information, or medical records. The research was based entirely on publicly available trial-level data that have been fully and irreversibly anonymized and do not contain any personally identifiable information. Therefore, no identification of individual participant information is possible from this work [66].

Results

Baseline Characteristics of Included Studies

The meta-analysis included 21,236 patients with mild cognitive impairment (MCI) and early and mild-to-moderate sporadic AD in 26 studies (19 ClinicalTrials.gov registrations of phase

II and III RCTs; Figure 1). Seven antiamyloid monoclonal antibodies were evaluated: bapineuzumab [17,18,22], gantenerumab [25,26], aducanumab [15], solanezumab [19,20], lecanemab [16,21], donanemab [23,24], and crenezumab [27,67]. The analyzed RCTs compared the interventions with placebo during 82.8 weeks on average (69-116 weeks), with acetylcholinesterase inhibitors and memantine allowed (alone or combined). Table 1 provides an overview of the baseline participant characteristics of the primary studies. Further baseline characteristics and the inclusion and exclusion criteria of the primary studies are presented in Multimedia Appendices 3 and 4, respectively. The results of the quality of report assessments (modified Jadad scale) and the specific reasons for the exclusion of each study can be found in Multimedia Appendices 5 and 6, respectively. Regarding the risk of bias, all studies were considered as having "low risk" or "raising some concerns" (Multimedia Appendix 7). For clinically relevant outcomes, summary of findings tables with rated certainty of evidence (vs placebo) according to the GRADE approach can be found in Figures S1-S8 in Multimedia Appendix 8. The complete GRADE table with certainty of evidence for each comparison within the NMA can be found in Multimedia Appendix 9.



Figure 1. PRISMA (Preferred Reporting Items for Systematic Reviews and Meta-Analyses) flow diagram. AD: Alzheimer disease.





Study and dose	Dosage	Patients (placebo vs intervention), n	AD ^a stage (MMSE ^b score for inclusion)
Salloway et al [22], 2009; high dose	1 mg/kg q13w ^c for 78 weeks	22 vs 26	Mild-moderate (16-26)
Salloway et al [18], 2014; Study 301; low dose	0.5 mg/kg q13w for 78 weeks	493 vs 314	Mild-moderate (16-26)
Salloway et al [18], 2014; Study 301; high dose	1 mg/kg q13w for 78 weeks	493 vs 307	Mild-moderate (16-26)
Salloway et al [18], 2014; Study 302; low dose	0.5 mg/kg q13w for 78 weeks	432 vs 658	Mild-moderate (16-26)
Doody et al [20], 2014; EXPEDITION 1	400 mg q4w for 78 weeks	506 vs 506	Mild-moderate (16-26)
Doody et al [20], 2014; EXPEDITION 2	400 mg q4w for 78 weeks	519 vs 521	Mild-moderate (16-26)
Vandenberghe et al [17], 2016; low dose	0.5 mg/kg q13w for 78 weeks	328 vs 328	Mild-moderate (16-26)
Vandenberghe et al [17], 2016; high dose	0.5 mg/kg q13w for 78 weeks	328 vs 253	Mild-moderate (16-26)
Vandenberghe et al [17], 2016; low dose	1 mg/kg q13w for 78 weeks	431 vs 650	Mild-moderate (16-26)
Honig et al [19], 2018; EXPEDITION 3	400 mg q4w for 76 weeks	1057 vs 1072	Mild (20-26)
Budd Haeberlein et al [15], 2022; EMERGE; low dose	Titrated to a target dose of 3 mg/kg ($\epsilon 4^+$) or 6 mg/kg ($\epsilon 4^-$) q4w over 76 weeks	548 vs 543	Early (MCI ^d and mild) (24-30)
Budd Haeberlein et al [15], 2022; EMERGE; high dose	Titrated to a target dose of 6 mg/kg (ϵ 4 ⁺) or 10 mg/kg (ϵ 4 ⁻) q4w over 76 weeks	548 vs 547	Early (MCI and mild) (24-30)
Budd Haeberlein et al [15], 2022; ENGAGE; low dose	Titrated to a target dose of 3 mg/kg ($\epsilon 4^+$) or 6 mg/kg ($\epsilon 4^-$) q4w over 76 weeks	545 vs 547	Early (MCI and mild) (24-30)
Budd Haeberlein et al [15], 2022; ENGAGE; high dose	Titrated to a target dose of 6 mg/kg ($\epsilon 4^+$) or 10 mg/kg ($\epsilon 4^-$) q4w over 76 weeks	545 vs 555	Early (MCI and mild) (24-30)
van Dyck et al [16], 2023; Clarity AD	10 mg/kg every 2 weeks over 78 weeks	875 vs 859	Early (MCI and mild) (20-26)
Swanson et al [21], 2021; high dose	10 mg/kg every 2 weeks over 78 weeks	238 vs 152	Early (MCI and mild) (22-30)
Swanson et al [21], 2021; low dose	10 mg/kg every month for 78 weeks	238 vs 246	Early (MCI and mild) (22-30)
Sims et al [24], 2023; TRAILBLAZER-ALZ 2 (pooled)	700 mg for the first 3 doses and 1400 mg thereafter q4w over 72 weeks	876 vs 860	Early (MCI and mild) (20-28)
Mintun et al [23], 2021; TRAILBLAZER-ALZ	700 mg for the first 3 doses and 1400 mg thereafter q4w over 72 weeks	126 vs 131	Early (MCI and mild) (20-28)
Bateman et al [25], 2023; GRADUATE I	Minimum 3 doses at each stage: starting at 120 mg q4w and increasing to 255 mg q4w, then to 510 mg q4w, and finally to 510 mg q2w	485 vs 499	Early (MCI and mild) (≥22)
Bateman et al [25], 2023; GRADUATE II	Minimum 3 doses at each stage: starting at 120 mg q4w and increasing to 255 mg q4w, then to 510 mg q4w, and finally to 510 mg q2w	477 vs 498	Early (MCI and mild) (≥22)
Ostrowitzki et al [27], 2022; CREAD	60 mg/kg q4w for up to 100 weeks	86 vs 80	Early (prodromal-mild) (≥20)
Salloway et al [67], 2018; BLAZE (pooled)	Part 1: 300 mg subcutaneous (q2w); Part 2: 15 mg/kg intravenous (q4w)	29 vs 62	Mild-moderate (18-26)
Ostrowitzki et al [26], 2017; SCarlet RoAD I	105 mg by subcutaneous injection (q4w) for 104 weeks or nearly 2 years	133 vs 271	Early (≥24)
Ostrowitzki et al [26], 2017; SCarlet RoAD II	225 mg by subcutaneous injection (q4w) for 104 weeks or nearly 2 years	133 vs 260	Early (≥24)

^aAD: Alzheimer disease.

XSL•FO RenderX

^bMMSE: Mini-Mental State Examination.

^cqXw: once every X weeks.

https://www.jmir.org/2025/1/e68454

^dMCI: mild cognitive impairment.

Primary Outcomes

Regarding the ADAS-Cog, 5 antiamyloid antibodies were significantly more effective than placebo in the frequentist NMA (Table 2; Figure S1 in Multimedia Appendix 8). These showed small effect sizes (donanemab > gantenerumab > lecanemab > aducanumab > solanezumab), with PIs for solanezumab including zero, suggesting a possible nonsignificant difference from placebo. The Bayesian NMA validated these findings, with strong support for the effects of donanemab, gantenerumab, lecanemab, and aducanumab, and a lack of strong evidence for solanezumab.

In the MMSE, donanemab and solanezumab were the only treatments superior to placebo, with small effect sizes in the frequentist NMA (Table 2; Figure S2 in Multimedia Appendix 8). Convincing evidence in the Bayesian NMA was found only for the effects of donanemab.

In the CDR-SB, both the frequentist and Bayesian NMAs conclusively showed that donanemab was significantly more effective than placebo and all the other antibodies. While aducanumab (P=.04) and lecanemab (P=.005) demonstrated relative benefits over placebo in the frequentist NMA with small effect sizes and CIs/PIs very close to zero (Table 2; Figure S3 in Multimedia Appendix 8), the Bayesian NMA results did not

provide conclusive support for these effects. In other words, while donanemab was superior to placebo and other antibodies on the CDR-SB, lecanemab and aducanumab could be associated with slightly higher benefits than placebo. However, we cannot confidently rule out the possibility of no significant difference.

Heterogeneity was nonsignificant for primary outcomes ($l^2=0\%$), and inconsistency was not assessed because the networks were lacking closed loops. No evidence of publication bias was found, with Egger test *P*-values of .55, .91, and .80 for the ADAS-Cog, MMSE, and CDR-SB, respectively. Funnel plots were symmetric for primary outcomes, and the imputation of potentially "missing studies" did not change the results. The Bayesian network meta-regression revealed no impact of study characteristics (number of study sites, age of participants, and baseline ADAS-Cog score) on the effect size estimates for primary outcomes.

Among all primary outcomes, donanemab outperformed other antibodies according to P-scores in the frequentist framework and SUCRA scores in the Bayesian framework. SUCRA rankings were generally robust to study exclusions (Multimedia Appendix 10), and high certainty (low information entropy) in ranking probabilities was found for the CDR-SB, where donanemab produced the clinically most relevant effects (Table 2; Figure 2).



Jeremic et al

Table 2.	Frequentist network	neta-analysis summar	y of findings: clini	cal benefits on cog	nitive and functional measures
		· · · · · · · · · · · · · · · · · · ·	_ · · · · · ·		

Measure and drug	RCT ^a and patient count	Relative effect, SMD ^b (95% CI)	Absolute effect, NNTB ^c (95% CI)	Ranking ^d , P-score; SUCRA (rank uncertainty)	Evidence certainty ^e	Comparison to placebo
ADAS-Cog ^f		·				
DON ^g	3 RCTs and 1469 patients	-0.22 (-0.32 to -0.12)	14 (9 to 26)	0.97; 96.3% (0.81)	Low ^h	Probably better
LEC ⁱ	3 RCTs and 2109 patients	-0.11 (-0.19 to -0.02)	29 (16 to 157)	0.67; 66.0% (0.94)	Very low ^j	Probably better
ADU ^k	4 RCTs and 2301 patients	-0.10 (-0.18 to -0.02)	31 (17 to 157)	0.65; 64.2% (0.95)	Very low ^j	Probably better
GAN ^l	3 RCTs and 4172 patients	-0.10 (-0.18 to -0.01)	33 (17 to 314)	0.62; 60.6% (0.93)	Very low ^j	Probably better
SOL ^m	3 RCTs and 4172 patients	-0.07 (-0.13 to -0.01)	47 (24 to 524)	0.48; 48.6% (0.93)	Very low ⁿ	Possibly better; no con- clusive difference
MMSE ⁰						
DON	3 RCTs and 1460 patients	0.14 (0.04 to 0.25)	21 (12 to 131)	0.92; 90.2% (0.53)	Very low ^h	Probably better
SOL	3 RCTs and 3821 patients	0.09 (0.02 to 0.15)	47 (24 to 524)	0.76; 73.8% (0.74)	Very low ⁿ	Possibly better; no con- clusive difference
CDR-SB ^p						
DON	3 RCTs and 1452 patients	-0.29 (-0.40 to -0.19)	10 (8 to 16)	1.00; 99.0% (0.27)	Low ^h	Probably better
LEC	3 RCTs and 2128 patients	-0.12 (-0.21 to -0.03)	25 (15 to 104)	0.75; 72.46% (0.81)	Very low ⁿ	Possibly better; no con- clusive difference
ADU	4 RCTs and 2301 patients	-0.10 (-0.20 to -0.002)	31 (15 to 767)	0.68; 67.5% (0.87)	Very low ⁿ	Possibly better; no con- clusive difference

^aRCT: randomized controlled trial.

^bSMD: standardized mean difference.

^cNNTB: number needed to treat for an additional beneficial outcome (assumed control-event rate of 0.25).

 d In the frequentist network meta-analysis, the treatments are ranked with P-scores that measure the mean extent of certainty that a given treatment is better than the competing treatments, ranging from 0 to 1 (higher is better). In the Bayesian framework, the surface under the cumulative ranking curve (SUCRA) score (0%-100%) shows the average percentage of treatments that are worse than a given treatment. The closer the SUCRA score is to 100%, the more likely it is that the treatment is the best option. Rank uncertainty is measured by the normalized Shannon entropy score of Bayesian ranking probabilities, ranging from 0 (absolute certainty) to 1 (absolute uncertainty).

^eGrades: High, we are very confident that the true effect size lies close to that of the estimated effect; Moderate, we are moderately confident in the estimate, and the true effect size is likely to be close to our estimate, with the possibility that it is substantially different; Low, our confidence in the effect estimate is limited, and the true effect may be substantially different from the estimate of the effect; Very low, we have very little confidence in the effect estimate, and the true effect is likely to be substantially different from the estimate of the effect.

^fADAS-Cog: Alzheimer Disease Assessment Scale-Cognitive Subscale (lower SMD is better).

^gDON: donanemab.

^hDue to small effect size, imprecision, and wide prediction and credible intervals.

ⁱLEC: lecanemab.

^JDue to small effect size, imprecision, and wide prediction and credible intervals very close to zero.

^kADU: aducanumab.

¹GAN: gantenerumab.

^mSOL: solanezumab.

ⁿIn the frequentist network meta-analysis, the SMDs were statistically significant in the ADAS-Cog (P=.03) and MMSE (P=.007) for solanezumab, and in the CDR-SB for lecanemab (P=.007) and aducanumab (P=.046), but with confidence and prediction intervals including zero or very close to zero and large NNTB values. This suggests that these effects are unlikely to be clinically relevant. In the Bayesian framework, the interventions could be associated with a slightly higher benefit than placebo for the respective outcomes, but with credible intervals including zero; therefore, we cannot confidently rule out the possibility of no significant difference.

^oMMSE: Mini-Mental State Examination (higher SMD is better).

^pCDR-SB: Clinical Dementia Rating Scale-Sum of Boxes (lower SMD is better).

XSL•FO

Jeremic et al

Figure 2. Surface under the cumulative ranking curve (SUCRA) rank-heat plot for clinically relevant results achieved by more than one intervention. Bolded and underlined SUCRA values indicate moderate-to-high certainty in the ranking probabilities, with normalized Shannon information entropy lower than 0.5. ADAS-Cog: Alzheimer Disease Assessment Scale-Cognitive Subscale; CDR-SB: Clinical Dementia Rating Scale-Sum of Boxes; MMSE: Mini-Mental State Examination. *Lack of data on the corresponding outcome within the circle.



Secondary Outcomes

Biomarkers of $A\beta$ and p-tau

Biomarker analyses (available on AlzMeta.app) revealed more promising results of the approved antibodies. Both the frequentist and Bayesian NMAs of the PET data conclusively showed that 4 treatments significantly reduced A β brain burden by large effect sizes (donanemab > gantenerumab > lecanemab > aducanumab). Biomarkers of A β_{1-42} in the CSF were improved by 4 antibodies in the frequentist NMA (crenezumab > aducanumab > lecanemab > gantenerumab; not reported for donanemab); however, the Bayesian NMA provided strong support only for the effects of crenezumab and aducanumab. The results of amyloid biomarkers (CSF and PET) were not correlated with the mean changes observed in the primary outcomes.

CSF p-tau was significantly improved with 3 antibodies (aducanumab > gantenerumab > lecanemab; not reported for donanemab); however, the Bayesian NMA validated only the effects of aducanumab versus placebo. CSF p-tau measures were positively correlated with the effect sizes on the ADAS-Cog (r=0.77; P=.03) and CDR-SB (r=0.77; P=.03), with almost identical correlation coefficients and P-values for aducanumab, lecanemab, bapineuzumab, gantenerumab, and crenezumab.

Safety Outcomes

Antiamyloid antibodies did not increase the risk of serious AEs. However, 4 interventions (gantenerumab > donanemab > lecanemab > aducanumab) were less tolerable than placebo since they increased treatment discontinuations due to AEs by large effect sizes (Table 3; Figure S4 in Multimedia Appendix 8). Crenezumab and solanezumab were the most tolerable and safest antibodies in terms of ARIA-E and ARIA-H. All other antibodies substantially increased the risk of total ARIA-E (Table 3; Figure S5 in Multimedia Appendix 8). Heterogeneity was not significant for the risks of ARIA-E (I^2 =33.7%) and treatment dropouts due to AEs (I^2 =0%).



Jeremic et al

Table 3. Bayesian network meta-analysis summary of findings: clinically relevant risks.

Adverse event and drug	RCT ^a and patient count	Relative risk, RR ^b (95% CI)	Absolute effects ^c (95% CI)		Ranking ^f , P-score; SU- CRA (rank uncertainty)	Evidence certainty ^g	Comparison to placebo
			NNTB ^d , val- ue (95% CI)	AR ^e per 1000, value (95% CI)			
ARIA-E ^h (AC	R ⁱ =0.01)						
LEC ^j	3 RCTs and 2454 patients	8.23 (4.33-15.64)	14 (7-31)	77 (42-143)	0.46; 44.94% (0.78)	Moderatek	Probably worse
GAN ^l	4 RCTs and 2736 patients	9.36 (5.70-15.37)	12 (7-22)	91 (56-143)	0.40; 39.31% (0.78)	Moderatek	Probably worse
ADU ^m	4 RCTs and 3249 patients	11.40 (6.94- 18.73)	10 (6-17)	112 (67-167)	0.27; 28.38 (0.79)	Moderatek	Probably worse
BAP ⁿ	6 RCTs and 4278 patients	12.97 (6.82- 24.65)	9 (5-18)	125 (67-200)	0.19; 18.38% (0.72)	Moderatek	Probably worse
DON ⁰	3 RCTs and 1981 patients	13.60 (7.58- 24.38)	8 (5-16)	125 (72-200)	0.16; 17.71% (0.71)	Moderatek	Probably worse
ARIA-H ^p (AC	R=0.10)						
GAN	3 RCTs and 1770 patients	1.70 (1.36-2.14)	15 (9-28)	167 (125-200)	0.59; 58.62% (0.85)	Moderatek	Probably worse
LEC	3 RCTs and 2454 patients	1.89 (1.48-2.41)	12 (8-21)	167 (143-200)	0.50; 49.56% (0.86)	Moderatek	Probably worse
DON	3 RCTs and 1981 patients	2.43 (2.00-2.95)	8 (6-11)	200 (167-250)	0.21; 28.57% (0.86)	Moderatek	Probably worse
ADU	4 RCTs and 3253 patients	3.39 (2.75-4.18)	5 (4-6)	334 (250-334)	0.10; 11.39% (0.82)	Moderatek	Probably worse
Dropouts (AC	R=0.05)						
BAP	7 RCTs and 4507 patients	1.24 (1.01-1.52)	82 (28-1450)	59 (50-72)	0.59; 59.29% (0.78)	Very low ^q	Possibly worse; no conclusive difference
ADU	4 RCTs and 3249 patients	1.65 (1.07-2.54)	31 (13-283)	77 (53-125)	0.43; 43.20% (0.83)	Low ^r	Probably worse
LEC	3 RCTs and 2454 patients	2.18 (1.54-3.07)	17 (10-37)	100 (77-143)	0.29; 28.85% (0.74)	Moderatek	Probably worse
DON	3 RCTs and 1981 patients	3.04 (2.19-4.23)	10 (7-17)	143 (100-200)	0.11; 11.29% (0.66)	Moderatek	Probably worse
GAN	4 RCTs and 2756 patients	3.44 (2.32-5.10)	5 (3-10)	250 (143-334)	0.05; 6.21% (0.38)	Moderatek	Probably worse
Infusion-related reactions (ACR=0.04)							
LEC	3 RCTs and 2454 patients	4.08 (2.87-5.80)	9 (6-14)	143 (112-200)	0.50; 49.69% (0.42)	Very low ^q	Possibly worse; no conclusive difference
DON	3 RCTs and 1981 patients	17.65 (6.71- 46.43)	2 (1-5)	500 (250-1000)	0.22; 23.29% (0.47)	Moderatek	Probably worse
BAP	1 RCT and 1121 pa- tients	68.56 (9.39- 500.85)	1 (1-3)	1000 (334-1000)	0.03; 6.17% (0.38)	Very low ⁸	Probably worse

^aRCT: randomized controlled trial.

^bRR: risk ratio.

^cFor safety outcomes, absolute effects are expressed as absolute risk per 1000 people and as number needed to treat for an additional beneficial outcome, with assumed control risks equal to the average control risk for each outcome, weighted by the number of participants that received placebo.

^dNNTB: number needed to treat for an additional beneficial outcome.

^eAR: absolute risk.

 $^{\rm f}$ In the Bayesian framework, the surface under the cumulative ranking curve (SUCRA) score shows the average percentage of treatments that are worse than a given treatment, which can range from 0% to 100%. The closer the SUCRA score is to 100%, the more likely it is that the treatment is the best option. Rank uncertainty is measured by the normalized Shannon entropy score of Bayesian ranking probabilities, ranging from 0 (absolute certainty) to 1 (absolute uncertainty).

 g Grades: High, we are very confident that the true effect size lies close to that of the estimated effect; Moderate, we are moderately confident in the estimate, and the true effect size is likely to be close to our estimate, with the possibility that it is substantially different; Low, our confidence in the effect estimate is limited, and the true effect may be substantially different from the estimate of the effect; Very low, we have very little confidence in the effect estimate, and the true effect is likely to be substantially different from the estimate of the effect.

^hARIA-E: amyloid-related imaging abnormalities in the form of vasogenic cerebral edema and sulcal effusion.

ⁱACR: assumed control risk.

^jLEC: lecanemab.

^kDue to a large magnitude of the effect size.

¹GAN: gantenerumab.

^mADU: aducanumab.

ⁿBAP: bapineuzumab.

^oDON: donanemab.

^pARIA-H: amyloid-related imaging abnormalities in form of cerebral microhemorrhage and local siderosis.

^qIn the frequentist network meta-analysis, statistically significant risk was found in the infusion-related reactions for lecanemab (P<.001) and the treatment discontinuations for bapineuzumab (P=.04), but with prediction intervals close to 1 or including 1 (for bapineuzumab), suggesting that these effects are unlikely to be clinically relevant. In the Bayesian framework, these treatments could be associated with a slightly higher risk than placebo on the respective outcomes, but with credible intervals including 1; thus, we cannot confidently rule out the possibility of no significant difference.

^rDue to wide credible intervals and imprecision.

^sDue to wide credible intervals, imprecision, and a small number of studies.

The risk of ARIA-H was significant for 4 antibodies (aducanumab > donanemab > lecanemab > gantenerumab) (Table 3; Figure S6 in Multimedia Appendix 8), demonstrated conclusively by the frequentist and Bayesian NMAs. These estimates were obtained after excluding 1 influential study of gantenerumab (GRADUATE II) [25] that induced heterogeneity in the full dataset analysis.

Additional safety analyses showed that anti-A β antibodies did not raise the risks of dizziness, arthralgia, diarrhea, or urinary infections more than placebo. No significant heterogeneity was found with all interventions analyzed, except for headaches (I^2 =69%) and urinary infections (I^2 =53.3%), and excluding influential bapineuzumab [17] studies significantly reduced heterogeneity for both outcomes, leading to more certain estimates. This revealed that the risk of headaches was higher with aducanumab, lecanemab, and donanemab in the frequentist NMA (Figures 2 and 3; Figure S7 in Multimedia Appendix 8). However, the Bayesian NMA did not provide conclusive evidence for these comparisons, and frequentist PIs included 1 for all drugs, except for aducanumab. In addition, we found no enhanced risks of nasopharyngitis; however, these events were not reported for donanemab and lecanemab, which were antibodies tested during the COVID-19 pandemic (Multimedia Appendix 11).



Jeremic et al

Figure 3. Number needed to treat (NNT) section in AlzMeta.app 2.0 showing the frequentist and Bayesian estimates of NNT for an additional beneficial (NNTB) and harmful outcome (NNTH) for statistically significant results. The estimates can be obtained over a wide range of placebo responses, including the rates of Alzheimer disease progression and adverse event risks expected in the control group. Similarly, the absolute risk per 1000 people can be calculated based on assumed control risk. Furthermore, the Bayesian network meta-analysis (NMA) can be performed with different choices of heterogeneity priors (Multimedia Appendix 2): (1) uninformative priors (with large variance), (2) half-normal priors for SD, with scale factors based on overall heterogeneity measured in the frequentist NMA, and (3) half-normal priors assuming 3 times greater heterogeneity.



Significant positive correlations were found between the sample sizes of the intervention and control groups and the occurrences of various treatment-related AEs and potential nocebo effects. These AEs included nausea (r>0.96), dizziness (r>0.89), headaches (r>0.87), serious AEs (r>0.76), and treatment or placebo discontinuations due to AEs (r>0.54). Additionally, when all interventions were analyzed, no correlation was observed between the total number of ARIA-E events and the sample size in the treatment group, while a moderate correlation was found in the control group (r=0.52; P=.03). The correlations between ARIA-E events and sample sizes became significantly stronger after excluding outlying solanezumab studies [19,20], with r=0.67 (P=.004) in the control group and r=0.73 (P=.002) in the treatment group. No significant correlations were identified between the primary outcomes and safety outcomes, including ARIA events.

The risk of ARIA-E was clinically relevant (Table 3; Figure 3) for 5 antibodies (donanemab > bapineuzumab > aducanumab > gantenerumab > lecanemab). Donanemab and bapineuzumab were the least safe treatments based on the P-score and SUCRA score (Figure 2). For all these antibodies, except for bapineuzumab, the risk of treatment discontinuation due to AEs was also clinically meaningful (Table 3; Figure 3). Gantenerumab was the least tolerable, followed by donanemab. Safety SUCRA rankings (Figure 2) remained moderately to highly stable to study exclusion, with 89% (17/19), 50% (8/16), and 82% (9/11) of studies not affecting any rank for tolerability, ARIA-E, and ARIA-H, respectively (Multimedia Appendix 10).

Tertiary Outcomes

Given the limited information about tertiary outcomes, our results indicate that bapineuzumab, donanemab, lecanemab, aducanumab, and gantenerumab achieved clinically relevant risks of ARIA-E in both *APOE*-ɛ4 carriers and noncarriers (available in AlzMeta.app). The risk was higher in the carrier population, as reflected by CIs and PIs and lower *P*-values within the frequentist NMA. The higher risk in carriers was particularly noticeable for aducanumab and gantenerumab (approximately 3 times greater risk) and bapineuzumab (approximately 2 times greater risk) and less pronounced for donanemab and lecanemab, where a lower number of events was reported in noncarriers, giving more uncertain estimates.

When compared to placebo, 3 antibodies (bapineuzumab > donanemab > lecanemab) significantly increased the risk of infusion-related reactions (Table 2; Figure 3; Figure S8 in Multimedia Appendix 8), which showed mostly mild to moderate severity. These risks were all clinically relevant. Lecanemab was a relatively safer treatment option when compared to donanemab and bapineuzumab (Figure 2), with moderate uncertainty in ranking probabilities and credible intervals including 1 in the Bayesian framework. However, these estimates should be interpreted with caution due to the small number of studies and treatments included in the comparisons, leading to poor robustness of the SUCRA rankings (Multimedia Appendix 10).

Donanemab increased the risk of nausea in 2 out of 3 treatment arms [23,24]. AR measures revealed that the risk of nausea was clinically relevant, but with broad CIs (Figure 3) and

overlapping prediction and credible intervals, suggesting considerable uncertainty in the estimates. Further analyses showed that the anti-A β antibodies did not increase the risk of fatigue, back pain, or upper respiratory infections more than placebo [24,25].

Sensitivity Analyses

After excluding influential studies (Multimedia Appendix 12), we found greater risks and benefits of high-dose lecanemab and aducanumab (in the controversial EMERGE study [15,68]) for safety outcomes (ARIA-E, ARIA-H, and headaches), cognitive/functional outcomes (ADAS-Cog and CDR-SB), and biomarker outcomes (CSF and PET). Moreover, the risk of ARIA-E was dose-dependent for bapineuzumab, with a greater risk associated with a high dose. For low-dose aducanumab, no significant risk was found for treatment discontinuations due to AEs.

The dose-dependent efficacy of aducanumab and lecanemab was clinically meaningful on the ADAS-Cog and CDR-SB, with greater absolute benefits of high-dose regimens. However, the intervention with donanemab was still the best treatment option, being almost 2 times more effective than high-dose aducanumab and lecanemab. Further sensitivity analysis showed that donanemab had consistently greater cognitive and functional effects in patients with low or medium tau load than in the high-tau population. Still, even upon removing the low or medium tau study, donanemab remained the best-ranked treatment, outperforming the other drugs on cognitive and functional measures.

Impact of the COVID-19 Pandemic

Among the studies included in this work, 2 gantenerumab studies [25], 1 lecanemab study [16], and 3 donanemab studies [23,24] were affected by the global outbreak of coronavirus, which caused delays in study visits and assessments, possible unblinding of some participants, and other difficulties, which have been summarized in Multimedia Appendix 11. The greatest impact was found in 2 phase III donanemab studies [24], with COVID-19 being the most commonly reported AE and being diagnosed in 16.0% (136/852) of patients in the control group and 17.6% (154/873) of patients in the group that received donanemab. A less severe impact was found in a lecanemab phase III study [16], with COVID-19 diagnosed in 7.1% (64/901) of patients in the control group and 6.7% (60/896) of patients in the intervention group. The reported impact of the pandemic was minimal in other studies.

Discussion

Principal Findings

The findings of this study support the effectiveness of donanemab and lecanemab, revealing small effect sizes for cognitive and functional outcomes, as well as moderate and large effect sizes for the CSF biomarkers of $A\beta$ and p-tau, respectively. Notably, donanemab demonstrated greater benefits in patients with lower tau burden, suggesting a more pronounced effect in the earlier stages of AD. At the same time, the risks of ARIA-E, ARIA-H, and infusion-related reactions were substantial, indicating greater harm than benefit for all approved

```
https://www.jmir.org/2025/1/e68454
```

antibodies. The risks for donanemab were similar and potentially worse when compared to lecanemab and aducanumab. Lecanemab was a slightly safer and more tolerable option than donanemab and aducanumab, with lower risks of ARIA events and infusion-related reactions.

Comparison to Prior Work

The findings presented here are consistent with our previous research [14] and other meta-analyses on anti-A β antibodies [8,9,35,69-72]. Based on the evidence so far, lecanemab and donanemab achieved statistical significance at attenuating cognitive and functional decline; however, the effect sizes observed in clinical trials were far below the MCID [1,35,69,71,72]. None of the studies reported so far found other clinically important results, such as a reduction in functional dependence, placement in memory care units or nursing homes, caregiver burden, or development of aggressive behaviors [69]. Although the cognitive and functional benefits of these drugs remain uncertain and show small effect sizes, they effectively demonstrated target engagement and disease-modifying potential by reducing A β load and improved multiple biomarker measures linked to AD pathology, including amyloid burden on PET and CSF A β_{40} , A β_{42} , and p-tau [16,24,71]. Therefore, direct comparative and longitudinal studies are required to fully disclose the impact of these disease-modifying effects over longer periods of time and better assess treatment-emergent risks in different populations. Exploratory post hoc modeling based on a phase III RCT of donanemab [24] suggested that $A\beta$ levels in treated patients would remain below the positivity threshold for nearly 4 years without treatment, although it is uncertain whether the MCID would be achieved in the long term on cognitive and functional outcomes, since the effect sizes obtained so far are very low.

Drawing from our findings and earlier research, treatment initiation in earlier AD stages is expected to be more beneficial [24,71]; however, prolonged monitoring is essential to determine whether these benefits translate into sustained clinical advantages over time and whether there is a cumulative benefit that meets or exceeds the MCID. For clinicians, the MCID depends on the detected change from the last evaluation; however, a measure obtained in this way does not account for cost-effectiveness and may be variously applied across different studies, medical centers, and health care systems. Moreover, MCIDs derived from population averages may not reflect the diversity of experiences among AD patients, and small cognitive improvements observed with anti-A β therapies might be highly meaningful for a patient or setting but negligible for another. Ultimately, multidisciplinary evaluation integrating clinical, patient-reported, and economic outcomes is necessary to determine the true value of these treatments [13,73]. While the MCID remains useful, it should be complemented by additional clinically relevant metrics, such as NNTs, under a wide range of assumptions regarding the baseline risks of AEs and AD progression [12,74-76] and cost-effectiveness, and further modified by personalized thresholds based on real-world data addressing the complexities of AD, quality of life, and the various needs of patients, caregivers, and health care systems [13].

Strengths and Limitations

One of the main strengths of our study is its capacity to provide real-time estimates of both relative and absolute risks and benefits of interventions within multiple datasets and dose regimens. These calculations are performed based on user-specific prior assumptions concerning heterogeneity and expected control responses (Figure 3), and the intervention rankings are accompanied by measures of their robustness and uncertainty. To the best of our knowledge, this approach was not used in previous studies, and it allows researchers and clinicians to incorporate their own expectations based on clinical expertise and novel evidence, ensuring that the findings remain relevant and adaptable to various clinical scenarios. We have combined the frequentist and Bayesian approaches to NMA, allowing users to easily obtain the results with different models and examine their discrepancies. The frequentist approach, while widely taught at all education levels and valued for offering straightforward interpretations, is often criticized for its reliance on fixed assumptions about data distributions and its inability to formally incorporate prior knowledge into the analysis. This can limit its applicability and generalizability in clinical contexts where data may be sparse or uncertain. In contrast, Bayesian methods provide a flexible framework that incorporates prior information, which can improve the robustness of estimates and more comprehensive quantification of uncertainty. While Bayesian methods have gained substantial momentum in recent decades of research, it remains less recognized than the frequentist approach and is often avoided due to perceived complexities or misconceptions. Researchers and clinicians may hesitate to engage with Bayesian methods due to computational demands or unfamiliarity with its principles and interpretation [57,75-79].

Our web application (available in English and Spanish upon peer review of this work) is very easy to use and provides extensive insights into the meta-analytic methodology, data structure, and valuable information that can serve educational purposes for anyone interested in meta-analysis and these therapies. It can be a useful tool for students, medical practitioners, and researchers to gain insights into meta-analysis techniques, including conventional pair-wise meta-analyses, and frequentist and Bayesian NMAs. Researchers could use it to facilitate their own analyses and compare results across different study types or assumptions, enabling more informed publication and policy recommendations. For health care professionals, the application provides a straightforward way to analyze clinical trial data and obtain user-defined drug effect estimates, enhancing decision-making for patient care. Furthermore, AlzMeta.app generates PIs for each outcome within the frequentist NMA. PIs account for uncertainty (heterogeneity) in the intervention estimates and aim to predict future individual observations. Therefore, when a new patient comes to the clinic, PIs (rather than CIs) should be used to predict treatment effects and recommend the optimal treatment [80,81]. Finally, the approach presented here can be easily updated and scaled up in the future by incorporating novel treatment comparisons and other findings relevant to researchers, clinicians, and the broader community. We will continue to add more information and update AlzMeta.app annually or when a

large amount of new clinical trial data and drug recommendations become available.

Several limitations of this work need to be acknowledged and further resolved. First, it relied on published trial-level data without incorporating individual patient data or longitudinal findings. Second, potential bias might be introduced in primary studies due to unblinding caused by the COVID-19 pandemic and frequent ARIAs. A recent simulation study estimated that 70%-100% of ARIAs in trials led to some degree of therapeutic insight [82]. Further adaptations in trial design and more detailed reporting and analysis of unblinding events are required [82]. Third, the patients in a donanemab phase III study [24] were stratified by their brain tau levels, which was not performed in other trials, and direct head-to-head comparisons between high-clearance antibodies have not been performed yet. Therefore, our NMA only included studies where treatments were compared with placebo. Finally, clinical experience with antiamyloid monoclonal antibodies is still in its early stages and primarily limited to clinical trial populations, which may not fully represent real-world practice in the broader symptomatic AD population. The trials performed so far lacked racial and ethnic diversity (Multimedia Appendix 3), which prevented us from determining whether the risks and benefits of these treatments may vary across different populations. Similarly, patients with abnormal MRI findings, cerebrovascular damage, major depression, schizophrenia, and other serious neuropsychiatric conditions (occurring among patients with AD) were excluded from the primary studies (Multimedia Appendix 4).

Future Directions

Further studies should integrate real-world data collected outside the controlled settings of clinical trials and include assessments on broader and more diverse populations. This will provide more robust findings and richer insights into how these treatments perform in everyday clinical practice, especially concerning their safety and long-term outcomes in different patient populations with multiple comorbidities. As the low tolerance and significant risks associated with ARIAs and infusion-related reactions loom over the approved anti-AB antibodies, future research is expected to support a better understanding of ARIA etiology, prevention, and management. When ARIAs were first observed in clinical trials with anti-A β vaccines [83] and antibodies [22], permanent discontinuation of therapy was recommended upon ARIA detection. The current practice has evolved, and most cases of asymptomatic ARIA-E meeting specific clinical and radiographic criteria can be managed with continued treatment, as ARIA-E events can resolve spontaneously with or without temporary interruption [84]. However, moderate or more severe ARIA-E cases with symptoms like headache, dizziness, or seizures, may require therapy interruption, hospitalization, enhanced monitoring, including brain imaging and electroencephalography, and administration of corticosteroids or anticonvulsants.

Careful dose titration can help mitigate the risk of ARIA-E events, and corticosteroid treatment can reduce inflammation and the radiographic severity of ARIA-E [72,84-86]. The risks can be further minimized by excluding individuals with more

than four microhemorrhages at baseline and by avoiding patients on antiplatelet or anticoagulant drugs. In any case, continuation of therapy heavily relies on accurate grading and monitoring of radiological findings. Therefore, radiologists play a crucial role in closely supervising patients for any ARIA event. Permanent cessation is recommended for 10 or more severe ARIA-H cases (even if asymptomatic, with or without ARIA-E), due to the irreversibility of ARIA-H findings [72,87,88].

ARIA events are linked to inflammation, blood-brain barrier disruption, and cerebrovascular impairment, which are already common in elderly people and patients with AD [89-91]. The ARIA risk is dose-dependent and influenced by factors such as active and passive anti-A β immunotherapy, anticoagulant use, APOE-£4 carriership, and a history of cerebrovascular events [87,92]. The APOE- ε 4 allele dose (homozygotes > heterozygotes) is the strongest risk factor for ARIA development, treatment discontinuation, and symptom manifestation, likely due to increased vascular AB burden and reduced cerebrovascular integrity [16,23,72,87,93]. With that in mind, future studies need to explore the complex interplay between ARIA and these risk factors. Mechanistic research could provide insights into the underlying biological pathways and factors that contribute to ARIA development, such as anti-A β immunization, baseline cerebrovascular integrity, and inflammatory responses, including complement cascade and microglia response [84,94]. Longitudinal studies should examine

how these factors influence ARIA evolution in diverse patient populations to account for variability in clinical presentations and responses to therapy. Likewise, animal models that mimic ARIA-like events could help in identifying molecular mechanisms, testing interventions, and developing predictive biomarkers for ARIA. These efforts will be crucial for improving patient selection, optimizing treatment protocols, and mitigating risks in the clinical use of anti-A β antibodies [86]. Furthermore, these studies may aid in the development of personalized approaches that integrate vascular, neuronal, glial, and inflammatory biomarkers with multi-target therapies to improve AD treatment.

In summary, our results demonstrate the small cognitive and functional benefits and substantial biomarker effects of donanemab and lecanemab, and the superiority of donanemab for cognitive and functional outcomes. Yet, these effects were achieved with significant trade-offs in the form of high risks of treatment dropout, infusion-related reactions, cerebral vasogenic edema or sulcal effusion (ARIA-E), and microbleeding and local siderosis (ARIA-H). The risks may outweigh the benefits, highlighting the need for safer therapies and further trials to better assess the risk-benefit profile and determine patient eligibility [90]. Personalized medicine and multi-target approaches will likely play an essential role in optimizing treatment for individual patients and advancing AD treatment [95].

Acknowledgments

This study was supported by grants PID2020-115823-GBI00 (MCIN/AEI/10.13039/501100011033), SBPLY/21/180501/000150 (JCCM/ERDF - A way of making Europe), and 2022-GRIN-34354 (UCLM/ERDF intramural funding) to LJ-D and JDN-L.

Data Availability

All data supporting this study are available on AlzMeta.app 2.0 [44].

Disclaimer

The conclusions drawn in this study are based on the specific datasets and methodologies employed and may not fully generalize to all clinical scenarios. The findings presented here cannot be used to supersede clinical trial data or influence the approval of any medication. Rather, this study provides an objective meta-analytic comparison (at the trial level) of different treatment options in phase II and III randomized placebo-controlled clinical trials. Further research is necessary to evaluate these treatments across broader populations and varied contexts.

Authors' Contributions

JDN-L and LJ-D were responsible for initial conceptualization, funding acquisition, supervision, and project administration. DJ was responsible for data curation, software development, formal analysis, visualization, and writing the original draft. DJ, JDN-L, and LJ-D contributed to writing – review and editing. All authors read and approved the final manuscript. JDN-L and LJ-D contributed equally and share last and corresponding authorship. JDN-L can be reached at juan.navarro@uclm.es.

Conflicts of Interest

None declared.

Multimedia Appendix 1

PRISMA-NMA checklist. [PDF File (Adobe PDF File), 210 KB-Multimedia Appendix 1]



Multimedia Appendix 2

Supplementary methods. [DOCX File, 41 KB-Multimedia Appendix 2]

Multimedia Appendix 3

Additional baseline participant characteristics. [DOCX File , 33 KB-Multimedia Appendix 3]

Multimedia Appendix 4

Inclusion and exclusion criteria of primary studies (clinical trials) included in this review. [DOCX File , 46 KB-Multimedia Appendix 4]

Multimedia Appendix 5

Quality of the included reports assessed by the modified Jadad scale with 9 items. [DOCX File , 28 KB-Multimedia Appendix 5]

Multimedia Appendix 6

Excluded studies and reasons for exclusion. [DOCX File, 111 KB-Multimedia Appendix 6]

Multimedia Appendix 7

Risk of bias. [PPTX File , 494 KB-Multimedia Appendix 7]

Multimedia Appendix 8

Summary of findings tables. [PPTX File , 12443 KB-Multimedia Appendix 8]

Multimedia Appendix 9

Summary of findings tables for each comparison within the network meta-analysis (GRADE). [DOCX File , 139 KB-Multimedia Appendix 9]

Multimedia Appendix 10

Surface under the cumulative ranking curve rank robustness. [DOCX File , 34 KB-Multimedia Appendix 10]

Multimedia Appendix 11

Reported impact of the COVID-19 pandemic on the primary studies. [DOCX File , 21 KB-Multimedia Appendix 11]

Multimedia Appendix 12

Influential studies and their impacts on heterogeneity. [DOCX File , 32 KB-Multimedia Appendix 12]

References

- Wessels AM, Dennehy EB, Dowsett SA, Dickson SP, Hendrix SB. Meaningful clinical changes in Alzheimer disease measured with the iADRS and illustrated using the donanemab TRAILBLAZER-ALZ study findings. Neurol Clin Pract. Apr 2023;13(2):e200127. [FREE Full text] [doi: 10.1212/CPJ.000000000200127] [Medline: 36891463]
- Thussu S, Naidu A, Manivannan S, Grossberg GT. Profiling aducanumab as a treatment option for Alzheimer's disease: an overview of efficacy, safety and tolerability. Expert Rev Neurother. Nov 2024;24(11):1045-1053. [doi: 10.1080/14737175.2024.2402058] [Medline: 39291991]

- 3. Belder CRS, Schott JM, Fox NC. Preparing for disease-modifying therapies in Alzheimer's disease. Lancet Neurol. Sep 2023;22(9):782-783. [doi: 10.1016/S1474-4422(23)00274-0] [Medline: 37463598]
- 4. Cummings J, Osse AML, Cammann D, Powell J, Chen J. Anti-amyloid monoclonal antibodies for the treatment of Alzheimer's disease. BioDrugs. Jan 13, 2024;38(1):5-22. [FREE Full text] [doi: 10.1007/s40259-023-00633-2] [Medline: 37955845]
- 5. Blair HA. Lecanemab in Alzheimer's disease: a profile of its use. Drugs Ther Perspect. Sep 20, 2024;40(10):393-400. [doi: 10.1007/s40267-024-01108-2]
- 6. Cooper C, Marshall CR, Schott JM, Banerjee S. Preparing for disease-modifying dementia therapies in the UK. Nat Rev Neurol. Nov 2024;20(11):641-642. [doi: 10.1038/s41582-024-01022-1] [Medline: 39289581]
- 7. Digma LA, Winer JR, Greicius MD. Substantial doubt remains about the efficacy of anti-amyloid antibodies. J Alzheimers Dis. 2024;97(2):567-572. [doi: <u>10.3233/JAD-231198</u>] [Medline: <u>38250779</u>]
- 8. Lacorte E, Ancidoni A, Zaccaria V, Remoli G, Tariciotti L, Bellomo G, et al. Safety and efficacy of monoclonal antibodies for Alzheimer's disease: a systematic review and meta-analysis of published and unpublished clinical trials. J Alzheimers Dis. 2022;87(1):101-129. [FREE Full text] [doi: 10.3233/JAD-220046] [Medline: 35275549]
- Ramanan VK, Armstrong MJ, Choudhury P, Coerver KA, Hamilton RH, Klein BC, et al. Antiamyloid monoclonal antibody therapy for Alzheimer disease: emerging issues in neurology. Neurology. Nov 07, 2023;101(19):842-852. [doi: 10.1212/WNL.000000000207757] [Medline: <u>37495380</u>]
- 10. Høilund-Carlsen P, Alavi A, Barrio JR, Castellani RJ, Costa T, Herrup K, et al. Donanemab, another anti-Alzheimer's drug with risk and uncertain benefit. Ageing Res Rev. Aug 2024;99:102348. [doi: <u>10.1016/j.arr.2024.102348</u>] [Medline: <u>38830549</u>]
- Morris GP, Clark IA, Vissel B. Questions concerning the role of amyloid-β in the definition, aetiology and diagnosis of Alzheimer's disease. Acta Neuropathol. Nov 2018;136(5):663-689. [FREE Full text] [doi: 10.1007/s00401-018-1918-8] [Medline: 30349969]
- 12. Goldberg TE, Lee S, Devanand DP, Schneider LS. Comparison of relative change with effect size metrics in Alzheimer's disease clinical trials. J Neurol Neurosurg Psychiatry. Dec 14, 2023;95(1):2-7. [doi: 10.1136/jnnp-2023-331941] [Medline: 37979967]
- Cummings J. Meaningful benefit and minimal clinically important difference (MCID) in Alzheimer's disease: Open peer commentary. Alzheimers Dement (N Y). 2023;9(3):e12411. [FREE Full text] [doi: 10.1002/trc2.12411] [Medline: 37521521]
- Jeremic D, Navarro-López J, Jiménez-Díaz L. Efficacy and safety of anti-amyloid-β monoclonal antibodies in current Alzheimer's disease phase III clinical trials: A systematic review and interactive web app-based meta-analysis. Ageing Res Rev. Sep 2023;90:102012. [doi: <u>10.1016/j.arr.2023.102012</u>] [Medline: <u>37423541</u>]
- 15. Budd Haeberlein S, Aisen PS, Barkhof F, Chalkias S, Chen T, Cohen S, et al. Two randomized phase 3 studies of aducanumab in early Alzheimer's disease. J Prev Alzheimers Dis. 2022;9(2):197-210. [doi: 10.14283/jpad.2022.30] [Medline: 35542991]
- van Dyck CH, Swanson CJ, Aisen P, Bateman RJ, Chen C, Gee M, et al. Lecanemab in early Alzheimer's disease. N Engl J Med. Jan 05, 2023;388(1):9-21. [doi: <u>10.1056/NEJMoa2212948</u>] [Medline: <u>36449413</u>]
- 17. Vandenberghe R, Rinne JO, Boada M, Katayama S, Scheltens P, Vellas B, et al. Bapineuzumab 30003001 Clinical Study Investigators. Bapineuzumab for mild to moderate Alzheimer's disease in two global, randomized, phase 3 trials. Alzheimers Res Ther. May 12, 2016;8(1):18. [FREE Full text] [doi: 10.1186/s13195-016-0189-7] [Medline: 27176461]
- Salloway S, Sperling R, Fox NC, Blennow K, Klunk W, Raskind M, et al. Bapineuzumab 301302 Clinical Trial Investigators. Two phase 3 trials of bapineuzumab in mild-to-moderate Alzheimer's disease. N Engl J Med. Jan 23, 2014;370(4):322-333.
 [FREE Full text] [doi: 10.1056/NEJMoa1304839] [Medline: 24450891]
- Honig LS, Vellas B, Woodward M, Boada M, Bullock R, Borrie M, et al. Trial of solanezumab for mild dementia due to Alzheimer's disease. N Engl J Med. Jan 25, 2018;378(4):321-330. [FREE Full text] [doi: 10.1056/NEJMoa1705971] [Medline: 29365294]
- Doody RS, Thomas RG, Farlow M, Iwatsubo T, Vellas B, Joffe S, Alzheimer's Disease Cooperative Study Steering Committee, et al. Solanezumab Study Group. Phase 3 trials of solanezumab for mild-to-moderate Alzheimer's disease. N Engl J Med. Jan 23, 2014;370(4):311-321. [doi: 10.1056/NEJMoa1312889] [Medline: 24450890]
- Swanson CJ, Zhang Y, Dhadda S, Wang J, Kaplow J, Lai RYK, et al. A randomized, double-blind, phase 2b proof-of-concept clinical trial in early Alzheimer's disease with lecanemab, an anti-Aβ protofibril antibody. Alzheimers Res Ther. Apr 17, 2021;13(1):80. [FREE Full text] [doi: 10.1186/s13195-021-00813-8] [Medline: 33865446]
- 22. Salloway S, Sperling R, Gilman S, Fox NC, Blennow K, Raskind M, et al. Bapineuzumab 201 Clinical Trial Investigators. A phase 2 multiple ascending dose trial of bapineuzumab in mild to moderate Alzheimer disease. Neurology. Dec 15, 2009;73(24):2061-2070. [FREE Full text] [doi: 10.1212/WNL.0b013e3181c67808] [Medline: 19923550]
- 23. Mintun MA, Lo AC, Duggan Evans C, Wessels AM, Ardayfio PA, Andersen SW, et al. Donanemab in early Alzheimer's disease. N Engl J Med. May 06, 2021;384(18):1691-1704. [doi: 10.1056/NEJMoa2100708] [Medline: 33720637]
- 24. Sims JR, Zimmer JA, Evans CD, Lu M, Ardayfio P, Sparks J, et al. TRAILBLAZER-ALZ 2 Investigators. Donanemab in early symptomatic Alzheimer disease: the TRAILBLAZER-ALZ 2 randomized clinical trial. JAMA. Aug 08, 2023;330(6):512-527. [FREE Full text] [doi: 10.1001/jama.2023.13239] [Medline: 37459141]
- Bateman RJ, Smith J, Donohue MC, Delmar P, Abbas R, Salloway S, et al. GRADUATE III Investigatorsthe Gantenerumab Study Group. Two phase 3 trials of gantenerumab in early Alzheimer's disease. N Engl J Med. Nov 16, 2023;389(20):1862-1876. [doi: 10.1056/NEJMoa2304430] [Medline: <u>37966285</u>]

- 26. Ostrowitzki S, Lasser RA, Dorflinger E, Scheltens P, Barkhof F, Nikolcheva T, et al. SCarlet RoAD Investigators. A phase III randomized trial of gantenerumab in prodromal Alzheimer's disease. Alzheimers Res Ther. Dec 08, 2017;9(1):95. [FREE Full text] [doi: 10.1186/s13195-017-0318-y] [Medline: 29221491]
- 27. Ostrowitzki S, Bittner T, Sink KM, Mackey H, Rabe C, Honig LS, et al. Evaluating the safety and efficacy of crenezumab vs placebo in adults with early Alzheimer disease: two phase 3 randomized placebo-controlled trials. JAMA Neurol. Nov 01, 2022;79(11):1113-1121. [FREE Full text] [doi: 10.1001/jamaneurol.2022.2909] [Medline: 36121669]
- DerSimonian R, Laird N. Meta-analysis in clinical trials revisited. Contemp Clin Trials. Nov 2015;45(Pt A):139-145. [FREE Full text] [doi: 10.1016/j.cct.2015.09.002] [Medline: 26343745]
- 29. Jackson D, Bowden J, Baker R. How does the DerSimonian and Laird procedure for random effects meta-analysis compare with its more efficient but harder to compute counterparts? Journal of Statistical Planning and Inference. Apr 2010;140(4):961-970. [FREE Full text] [doi: 10.1016/j.jspi.2009.09.017]
- 30. Balduzzi S, Rücker G, Nikolakopoulou A, Papakonstantinou T, Salanti G, Efthimiou O, et al. netmeta: an R package for network meta-analysis using frequentist methods. J Stat Soft. 2023;106(2):1-40. [doi: 10.18637/jss.v106.i02]
- 31. van Valkenhoef G, Lu G, de Brock B, Hillege H, Ades AE, Welton NJ. Automating network meta-analysis. Res Synth Methods. Dec 2012;3(4):285-299. [doi: 10.1002/jrsm.1054] [Medline: 26053422]
- 32. Hutton B, Salanti G, Caldwell DM, Chaimani A, Schmid CH, Cameron C, et al. The PRISMA extension statement for reporting of systematic reviews incorporating network meta-analyses of health care interventions: checklist and explanations. Ann Intern Med. Jun 02, 2015;162(11):777-784. [FREE Full text] [doi: 10.7326/M14-2385] [Medline: 26030634]
- 33. GRADE Handbook: Introduction to GRADE Handbook. GRADE Pro. URL: <u>https://gdt.gradepro.org/app/handbook/</u> handbook.html [accessed 2025-03-07]
- Jadad AR, Moore RA, Carroll D, Jenkinson C, Reynolds DJ, Gavaghan DJ, et al. Assessing the quality of reports of randomized clinical trials: is blinding necessary? Control Clin Trials. Feb 1996;17(1):1-12. [doi: 10.1016/0197-2456(95)00134-4] [Medline: <u>8721797</u>]
- 35. Avgerinos KI, Ferrucci L, Kapogiannis D. Effects of monoclonal antibodies against amyloid-β on clinical and biomarker outcomes and adverse event risks: A systematic review and meta-analysis of phase III RCTs in Alzheimer's disease. Ageing Res Rev. Jul 2021;68:101339. [FREE Full text] [doi: 10.1016/j.arr.2021.101339] [Medline: 33831607]
- 36. Borenstein M, Hedges L, Higgins J, Rothstein H. Introduction to Meta Analysis. Hoboken, NJ. John Wiley & Sons; 2009.
- 37. Harrer M, Cuijpers P, Furukawa T, Ebert D. dmetar: Companion R Package For The Guide 'Doing Meta-Analysis in R'. R package version 0.1.0. dmetar. 2019. URL: <u>http://dmetar.protectlab.org/</u> [accessed 2025-03-07]
- 38. Cumpston M, Li T, Page MJ, Chandler J, Welch VA, Higgins JP, et al. Updated guidance for trusted systematic reviews: a new edition of the Cochrane Handbook for Systematic Reviews of Interventions. Cochrane Database Syst Rev. Oct 03, 2019;10(10):ED000142. [FREE Full text] [doi: 10.1002/14651858.ED000142] [Medline: 31643080]
- 39. Cochrane Handbook for Systematic Reviews of Interventions Version 6.2, 2021. Cochrane. 2021. URL: <u>https://training.cochrane.org/handbook/archive/v6.2</u> [accessed 2025-03-07]
- 40. Higgins J, Thomas J, Chandler J, Cumpston M, Li T, Page M, et al. Cochrane Handbook for Systematic Reviews of Interventions. Hoboken, NJ. John Wiley & Sons; 2019.
- 41. Furukawa TA. From effect size into number needed to treat. The Lancet. May 1999;353(9165):1680. [doi: 10.1016/s0140-6736(99)01163-0]
- 42. Furukawa TA, Leucht S. How to obtain NNT from Cohen's d: comparison of two methods. PLoS One. Apr 27, 2011;6(4):e19070. [FREE Full text] [doi: 10.1371/journal.pone.0019070] [Medline: 21556361]
- 43. Schünemann H, Vist G, Higgins J, Santesso N, Deeks J, Glasziou P, et al. Interpreting results and drawing conclusions. In: Cochrane Handbook for Systematic Reviews of Interventions, Second Edition. Hoboken, NJ. John Wiley & Sons; 2019:403-431.
- 44. AlzMeta.app. URL: https://alzmetaapp.shinyapps.io/ALZMETA_APP_2/ [accessed 2025-03-07]
- 45. Deeks J, Higgins J, Altman D. Analysing data and undertaking meta-analyses. In: Cochrane Handbook for Systematic Reviews of Interventions, Second Edition. Hoboken, NJ. John Wiley & Sons; 2019:241-284.
- 46. Rücker G, Schwarzer G. Ranking treatments in frequentist network meta-analysis works without resampling methods. BMC Med Res Methodol. Jul 31, 2015;15:58. [FREE Full text] [doi: 10.1186/s12874-015-0060-8] [Medline: 26227148]
- 47. Borenstein M, Hedges L, Higgins J, Rothstein H. Identifying and Quantifying Heterogeneity. In: Introduction to Meta Analysis. Hoboken, NJ. John Wiley & Sons; 2009:107-125.
- 48. McKenzie J, Brennan S, Ryan R, Thomson H, Johnston R, Thomas J. Defining the criteria for including studies and how they will be grouped for the synthesis. In: Cochrane Handbook for Systematic Reviews of Interventions, Second Edition. Hoboken, NJ. John Wiley & Sons; 2019:33-65.
- Guyatt GH, Oxman AD, Kunz R, Woodcock J, Brozek J, Helfand M, et al. GRADE Working Group. GRADE guidelines: 8. Rating the quality of evidence--indirectness. J Clin Epidemiol. Dec 2011;64(12):1303-1310. [doi: 10.1016/j.jclinepi.2011.04.014] [Medline: 21802903]
- Guyatt GH, Oxman AD, Kunz R, Woodcock J, Brozek J, Helfand M, et al. GRADE Working Group. GRADE guidelines:
 7. Rating the quality of evidence--inconsistency. J Clin Epidemiol. Dec 2011;64(12):1294-1302. [doi: 10.1016/j.jclinepi.2011.03.017] [Medline: 21803546]

- 51. van Valkenhoef G, Kuiper J. gemtc: Network Meta-Analysis Using Bayesian Methods, version 1.0–1, from CRAN. rdrr. URL: <u>https://rdrr.io/cran/gemtc/</u> [accessed 2025-03-07]
- Salanti G, Nikolakopoulou A, Efthimiou O, Mavridis D, Egger M, White IR. Introducing the treatment hierarchy question in network meta-analysis. Am J Epidemiol. Mar 24, 2022;191(5):930-938. [FREE Full text] [doi: 10.1093/aje/kwab278] [Medline: 35146500]
- Daly CH, Neupane B, Beyene J, Thabane L, Straus SE, Hamid JS. Empirical evaluation of SUCRA-based treatment ranks in network meta-analysis: quantifying robustness using Cohen's kappa. BMJ Open. Sep 05, 2019;9(9):e024625. [FREE Full text] [doi: 10.1136/bmjopen-2018-024625] [Medline: 31492773]
- 54. McHugh ML. Interrater reliability: the kappa statistic. Biochem Med (Zagreb). 2012;22(3):276-282. [FREE Full text] [Medline: 23092060]
- 55. Salanti G, Ades AE, Ioannidis JPA. Graphical methods and numerical summaries for presenting results from multiple-treatment meta-analysis: an overview and tutorial. J Clin Epidemiol. Feb 2011;64(2):163-171. [doi: 10.1016/j.jclinepi.2010.03.016] [Medline: 20688472]
- 56. Wu Y, Shih M, Tu Y. Using normalized entropy to measure uncertainty of rankings for network meta-analyses. Med Decis Making. Mar 23, 2021;41(6):706-713. [doi: 10.1177/0272989x21999023]
- 57. Harrer M, Cuijpers P, Furukawa T, Ebert D. Doing Meta-Analysis with R: A Hands-On Guide. Boca Raton, FL. Chapman and Hall/CRC; 2021.
- 58. Olkin I, Dahabreh IJ, Trikalinos TA. GOSH a graphical display of study heterogeneity. Res Synth Methods. Sep 11, 2012;3(3):214-223. [FREE Full text] [doi: 10.1002/jrsm.1053] [Medline: 26062164]
- Papakonstantinou T, Nikolakopoulou A, Rücker G, Chaimani A, Schwarzer G, Egger M, et al. Estimating the contribution of studies in network meta-analysis: paths, flows and streams. F1000Res. 2018;7:610. [FREE Full text] [doi: 10.12688/f1000research.14770.3] [Medline: 30338058]
- 60. Rücker G, Nikolakopoulou A, Papakonstantinou T, Salanti G, Riley RD, Schwarzer G. The statistical importance of a study for a network meta-analysis estimate. BMC Med Res Methodol. Jul 14, 2020;20(1):190. [FREE Full text] [doi: 10.1186/s12874-020-01075-y] [Medline: 32664867]
- 61. Sterne JAC, Savović J, Page MJ, Elbers RG, Blencowe NS, Boutron I, et al. RoB 2: a revised tool for assessing risk of bias in randomised trials. BMJ. Aug 28, 2019;366:14898. [FREE Full text] [doi: 10.1136/bmj.14898] [Medline: 31462531]
- 62. Higgins J, Savović J, Page M, Elbers R, Sterne J. Assessing risk of bias in a randomized trial. In: Cochrane Handbook for Systematic Reviews of Interventions, Second Edition. Hoboken, NJ. John Wiley & Sons; 2019:205-228.
- Sterne JAC, Sutton AJ, Ioannidis JPA, Terrin N, Jones DR, Lau J, et al. Recommendations for examining and interpreting funnel plot asymmetry in meta-analyses of randomised controlled trials. BMJ. Jul 22, 2011;343:d4002. [doi: 10.1136/bmj.d4002] [Medline: 21784880]
- 64. Egger M, Davey Smith G, Schneider M, Minder C. Bias in meta-analysis detected by a simple, graphical test. BMJ. Sep 13, 1997;315(7109):629-634. [FREE Full text] [doi: 10.1136/bmj.315.7109.629] [Medline: 9310563]
- 65. Package 'metafor'. CRAN. URL: <u>https://cran.r-project.org/web/packages/metafor/metafor.pdf</u> [accessed 2025-03-07]
- 66. Ethics and data protection. European Commission. URL: <u>https://commission.europa.eu/system/files/2020-06/5.</u> <u>h2020 ethics and data protection 0.pdf</u> [accessed 2025-03-07]
- 67. Salloway S, Honigberg LA, Cho W, Ward M, Friesenhahn M, Brunstein F, et al. Amyloid positron emission tomography and cerebrospinal fluid results from a crenezumab anti-amyloid-beta antibody double-blind, placebo-controlled, randomized phase II study in mild-to-moderate Alzheimer's disease (BLAZE). Alzheimers Res Ther. Sep 19, 2018;10(1):96. [FREE Full text] [doi: 10.1186/s13195-018-0424-5] [Medline: 30231896]
- 68. Kuller LH, Lopez OL. ENGAGE and EMERGE: Truth and consequences? Alzheimers Dement. Apr 2021;17(4):692-695. [FREE Full text] [doi: 10.1002/alz.12286] [Medline: 33656288]
- 69. Ebell MH, Barry HC, Baduni K, Grasso G. Clinically important benefits and harms of monoclonal antibodies targeting amyloid for the treatment of Alzheimer disease: a systematic review and meta-analysis. Ann Fam Med. 2024;22(1):50-62. [FREE Full text] [doi: 10.1370/afm.3050] [Medline: 38253509]
- Qiao Y, Gu J, Yu M, Chi Y, Ma Y. Comparative efficacy and safety of monoclonal antibodies for cognitive decline in patients with Alzheimer's disease: a systematic review and network meta-analysis. CNS Drugs. Mar 2024;38(3):169-192. [doi: 10.1007/s40263-024-01067-2] [Medline: 38429615]
- 71. Avgerinos KI, Manolopoulos A, Ferrucci L, Kapogiannis D. Critical assessment of anti-amyloid-β monoclonal antibodies effects in Alzheimer's disease: a systematic review and meta-analysis highlighting target engagement and clinical meaningfulness. Sci Rep. Oct 28, 2024;14(1):25741. [FREE Full text] [doi: 10.1038/s41598-024-75204-8] [Medline: 39468148]
- 72. Withington CG, Turner RS. Amyloid-related imaging abnormalities with anti-amyloid antibodies for the treatment of dementia due to Alzheimer's disease. Front Neurol. 2022;13:862369. [FREE Full text] [doi: 10.3389/fneur.2022.862369] [Medline: 35401412]
- 73. Muir RT, Hill MD, Black SE, Smith EE. Minimal clinically important difference in Alzheimer's disease: Rapid review. Alzheimers Dement. May 2024;20(5):3352-3363. [FREE Full text] [doi: 10.1002/alz.13770] [Medline: 38561021]

- 74. Mendes D, Alves C, Batel-Marques F. Number needed to treat (NNT) in clinical literature: an appraisal. BMC Med. Jun 01, 2017;15(1):112. [FREE Full text] [doi: 10.1186/s12916-017-0875-8] [Medline: 28571585]
- 75. Furukawa TA, Guyatt GH, Griffith LE. Can we individualize the 'number needed to treat'? An empirical study of summary effect measures in meta-analyses. Int J Epidemiol. Feb 2002;31(1):72-76. [doi: 10.1093/ije/31.1.72] [Medline: 11914297]
- Vancak V, Goldberg Y, Levine SZ. Guidelines to understand and compute the number needed to treat. Evid Based Ment Health. Nov 2021;24(4):131-136. [FREE Full text] [doi: 10.1136/ebmental-2020-300232] [Medline: 33619181]
- Muehlemann N, Zhou T, Mukherjee R, Hossain MI, Roychoudhury S, Russek-Cohen E. A tutorial on modern Bayesian methods in clinical trials. Ther Innov Regul Sci. May 2023;57(3):402-416. [FREE Full text] [doi: 10.1007/s43441-023-00515-3] [Medline: <u>37081374</u>]
- 78. König C, van de Schoot R. Bayesian statistics in educational research: a look at the current state of affairs. Educational Review. Jul 25, 2017;70(4):486-509. [doi: 10.1080/00131911.2017.1350636]
- 79. Medical Outreach Subteam of the Drug Information Association Bayesian Scientific Working Group, Clark J, Muhlemann N, Natanegara F, Hartley A, Wenkert D, et al. Why are not there more Bayesian clinical trials? perceived barriers and educational preferences among medical researchers involved in drug development. Ther Innov Regul Sci. May 2023;57(3):417-425. [FREE Full text] [doi: 10.1007/s43441-021-00357-x] [Medline: 34978048]
- 80. IntHout J, Ioannidis JPA, Rovers MM, Goeman JJ. Plea for routinely presenting prediction intervals in meta-analysis. BMJ Open. Jul 12, 2016;6(7):e010247. [FREE Full text] [doi: 10.1136/bmjopen-2015-010247] [Medline: 27406637]
- Lin L. Use of prediction intervals in network meta-analysis. JAMA Netw Open. Aug 02, 2019;2(8):e199735. [FREE Full text] [doi: 10.1001/jamanetworkopen.2019.9735] [Medline: 31433478]
- 82. Wolters FJ, Labrecque JA. Potential impact of unblinding on observed treatment effects in Alzheimer's disease trials. Alzheimers Dement. Apr 2024;20(4):3119-3125. [FREE Full text] [doi: 10.1002/alz.13690] [Medline: 38380503]
- Mantile F, Prisco A. Vaccination against β-amyloid as a strategy for the prevention of Alzheimer's disease. Biology (Basel). Nov 27, 2020;9(12):425. [FREE Full text] [doi: 10.3390/biology9120425] [Medline: 33260956]
- 84. Hampel H, Elhage A, Cho M, Apostolova LG, Nicoll JAR, Atri A. Amyloid-related imaging abnormalities (ARIA): radiological, biological and clinical characteristics. Brain. Nov 02, 2023;146(11):4414-4424. [FREE Full text] [doi: 10.1093/brain/awad188] [Medline: <u>37280110</u>]
- 85. Iwatsubo T. Development of disease-modifying therapies against Alzheimer's disease. Psychiatry Clin Neurosci. Sep 2024;78(9):491-494. [doi: 10.1111/pcn.13681] [Medline: 38842037]
- 86. Sperling RA, Jack CR, Black SE, Frosch MP, Greenberg SM, Hyman BT, et al. Amyloid-related imaging abnormalities in amyloid-modifying therapeutic trials: recommendations from the Alzheimer's Association Research Roundtable Workgroup. Alzheimers Dement. Jul 2011;7(4):367-385. [FREE Full text] [doi: 10.1016/j.jalz.2011.05.2351] [Medline: 21784348]
- 87. Cogswell PM, Barakos JA, Barkhof F, Benzinger TS, Jack CR, Poussaint TY, et al. Amyloid-related imaging abnormalities with emerging Alzheimer disease therapeutics: detection and reporting recommendations for clinical practice. AJNR Am J Neuroradiol. Sep 2022;43(9):E19-E35. [FREE Full text] [doi: 10.3174/ajnr.A7586] [Medline: 35953274]
- 88. Agarwal A, Gupta V, Brahmbhatt P, Desai A, Vibhute P, Joseph-Mathurin N, et al. Amyloid-related imaging abnormalities in Alzheimer disease treated with anti-amyloid-β therapy. Radiographics. Sep 2023;43(9):e230009. [FREE Full text] [doi: 10.1148/rg.230009] [Medline: <u>37651273</u>]
- 89. Biessels GJ. Alzheimer's disease, cerebrovascular disease and dementia: lump, split or integrate? Brain. Aug 27, 2022;145(8):2632-2634. [FREE Full text] [doi: 10.1093/brain/awac228] [Medline: 35848864]
- Cummings J, Aisen P, Apostolova LG, Atri A, Salloway S, Weiner M. Aducanumab: appropriate use recommendations. J Prev Alzheimers Dis. 2021;8(4):398-410. [FREE Full text] [doi: 10.14283/jpad.2021.41] [Medline: 34585212]
- 91. Rafii M. Editorial: Appropriate use recommendations for lecanemab. J Prev Alzheimers Dis. 2023;10(3):356. [doi: 10.14283/jpad.2023.34] [Medline: 37357273]
- 92. Doran SJ, Sawyer RP. Risk factors in developing amyloid related imaging abnormalities (ARIA) and clinical implications. Front Neurosci. 2024;18:1326784. [FREE Full text] [doi: 10.3389/fnins.2024.1326784] [Medline: 38312931]
- 93. Salloway S, Chalkias S, Barkhof F, Burkett P, Barakos J, Purcell D, et al. Amyloid-related imaging abnormalities in 2 phase 3 studies evaluating aducanumab in patients with early Alzheimer disease. JAMA Neurol. Jan 01, 2022;79(1):13-21. [FREE Full text] [doi: 10.1001/jamaneurol.2021.4161] [Medline: 34807243]
- 94. Söderberg L, Johannesson M, Gkanatsiou E, Nygren P, Fritz N, Zachrisson O, et al. Amyloid-beta antibody binding to cerebral amyloid angiopathy fibrils and risk for amyloid-related imaging abnormalities. Sci Rep. May 13, 2024;14(1):10868. [FREE Full text] [doi: 10.1038/s41598-024-61691-2] [Medline: 38740836]
- 95. Toups K, Hathaway A, Gordon D, Chung H, Raji C, Boyd A, et al. Precision medicine approach to Alzheimer's disease: successful pilot project. Journal of Alzheimer's Disease. Jul 04, 2022;88(4):1411-1421. [doi: <u>10.3233/jad-215707</u>]

Abbreviations

Aβ: amyloid βAD: Alzheimer diseaseADAS-Cog: AD Assessment Scale-Cognitive Subscale



AE: adverse event **AR:** absolute risk **ARIA:** amyloid-related imaging abnormality **ARIA-E:** amyloid-related imaging abnormalities in the form of vasogenic cerebral edema and sulcal effusion ARIA-H: amyloid-related imaging abnormalities in the form of microhemorrhage and superficial siderosis CDR-SB: Clinical Dementia Rating Scale-Sum of Boxes **CSF:** cerebrospinal fluid **GRADE:** Grading of Recommendations, Assessment, Development, and Evaluation MCID: minimal clinically important difference MCMC: Markov Chain Monte Carlo **MMSE:** Mini-Mental State Examination **NMA:** network meta-analysis **NNT:** number needed to treat **PET:** positron emission tomography **PI:** prediction interval PRISMA-NMA: Preferred Reporting Items for Systematic Reviews and Meta-Analyses extension for reporting of systematic reviews with network meta-analysis **RCT:** randomized placebo-controlled clinical trial **SMD:** standardized mean difference SUCRA: surface under the cumulative ranking curve

Edited by A Schwartz; submitted 06.11.24; peer-reviewed by TF Ojo, DB Foroutan, M Gasmi; comments to author 11.12.24; revised version received 09.01.25; accepted 21.02.25; published 07.04.25

<u>Please cite as:</u> Jeremic D, Navarro-Lopez JD, Jimenez-Diaz L Clinical Benefits and Risks of Antiamyloid Antibodies in Sporadic Alzheimer Disease: Systematic Review and Network Meta-Analysis With a Web Application J Med Internet Res 2025;27:e68454 URL: <u>https://www.jmir.org/2025/1/e68454</u> doi: <u>10.2196/68454</u> PMID:

©Danko Jeremic, Juan D Navarro-Lopez, Lydia Jimenez-Diaz. Originally published in the Journal of Medical Internet Research (https://www.jmir.org), 07.04.2025. This is an open-access article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work, first published in the Journal of Medical Internet Research (ISSN 1438-8871), is properly cited. The complete bibliographic information, a link to the original publication on https://www.jmir.org/, as well as this copyright and license information must be included.

